Author's response to reviews

Title: Epidemiology of cardiovascular risk factors in Greece; aims, design and baseline characteristics of the ATTICA study

Authors:

Dr Christos Pitsavos (d.b.panagiotakos@usa.net)
Dr Demosthenes B Panagiotakos (d.b.panagiotakos@usa.net)
Dr Christina Chrysohoou (d.b.panagiotakos@usa.net)
Dr John Skoumas (d.b.panagiotakos@usa.net)
Dr Christodoulos Stefanadis (d.b.panagiotakos@usa.net)

Version: 5 Date: 30 Sep 2003

Dear Editor,

Thank you for your latest email. We have made the requested changes in the Abstract. Also please note that we have removed one of the authors (J Skoumas) and we have placed his name in the end of the article.

Yours truly

Demosthenes B Panagiotakos